Author's response to reviews

Title: Glucagon like Peptide Analogues for Type 2 Diabetes Mellitus: Systematic Review and Meta-analysis

Authors:

Deepson S Shyangdan (d.s.shyangdan@abdn.ac.uk)
Pamela L Royle (p.royle@abdn.ac.uk)
Christine Clar (clar@cc-archie.de)
Pawana Sharma (p.sharma@abdn.ac.uk)
Norman R Waugh (n.r.waugh@abdn.ac.uk)

Version: 4 Date: 3 December 2010

Author's response to reviews: see over
Dear Editors,

Please find attached a revised version of our paper. Our changes are tabulated below.

**Major revisions**
- The names of authors have been written exactly as they are in the submission system, both in style and order.
- On the title page, we have added a statement ‘All authors contributed equally to this work’.
- On the title page, we have added a statement ‘All authors have read and approved the final manuscript’.
- Full address of corresponding author.
- All the highlighted sections in the manuscript have been removed.
- Table 1 and Table 2 have been added in the manuscript following the references. Table 3 and Table 4 have been cited in the manuscript as additional files.

**Minor revisions**
- Titles of the figures have been removed.
- Manuscript has been thoroughly checked for typographical errors. This is the final version of our manuscript.

**Prepared by:**

Deepson Shyangdan, research assistant and Pamela Royle, research fellow, Aberdeen Health Technology Assessment Group, and approved by all authors.